Author's response to reviews

Title: Factors Associated with Utilization of Long Acting and Permanent Contraceptive Methods among Married Women of Reproductive Age in Mekelle Town, Tigray Region North Ethiopia

Authors:

Mussie Alemayehu (mossalex75@gmail.com)
Tefera Belachew (tefera_belachew@yahoo.com)
Tizta Tilahun (tiztatilahun@yahoo.com)

Version: 8 Date: 21 January 2012

Author's response to reviews: see over
Formatting changes to be made:

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is "First name Initial Last name" (e.g. Joe F Bloggs)

Competing interests: Please include a "Competing interests" section between the Conclusions and Authors' contributions. If there are none to declare, please write 'The authors declare that they have no competing interests'. Please see here for a full list of questions to consider when writing your competing interests statement-

Response: We included it

References: please provide a functioning link to website 4

Response: Now is the web is functional

References: references 15 and 16, Unpublished. Articles that have not been accepted for publication should not be cited in the reference list, unless they are publicly available as a pre-print. Instead, please refer to them in the text in the following way: (Authors: Title of manuscript, submitted).

Response: We put it accordingly

15. (Getachew M: Assessment of the prevalence and factors affecting use of permanent and long acting contraceptive methods in Jinka town, south Omo Zone, SNNPR, Ethiopia, submitted to Addis Ababa University)
16. (Temesgen A: Assessment of the prevalence and factors influencing the utilization of long acting and permanent contraceptive method in Butajira town, Gurage zone, SNNP, Ethiopia, submitted to Addis Ababa University)

References: Please make sure references are cited in numerical order. We notice references 9 and 18 are not cited within the main text, and references 15, 19 and 24 are cited out of order. Please make sure references continue to be cited in numerical order once corrections have been made.

Response: reference 15 cancelled from the list, Reference 19 changed to 16 and Reference 24 modified to 20

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Title: please use sentence case in the title, both in the manuscript file and the submission system. Capitalise only the first word and proper nouns, and do not include full stops/periods. For example, "The Title of My Prestigious Publication: A Scientific Study." should be changed to "The title of my prestigious publication: a scientific study"

Response: corrected accordingly
Highlighting/tracking: Please remove all highlighting and tracking from the manuscript, and change any coloured text to black

Response: Ok done

Tables: please note that we are unable to correctly display merged cells where the merged cell crosses rows: please re-layout your table without these merged elements. Please also change all coloured elements to black

Response: Ok done

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Response: Ok we have done thoroughly